Author's response to reviews

Title: Association between Paraoxonase gene and stroke in the Han Chinese Population

Authors:

Guojun Zhang (zguoj1018@yahoo.com.cn)
Wenjin Li (wenjinliwl@gmail.com)
Zhiqiang Li (lizqsitu@gmail.com)
Hong Lv (lvhonghong@yahoo.com.cn)
Yonghong Ren (yhren@capitalbio.com)
Ruimin Ma (ma.ruimin@yahoo.com.cn)
Xiaohong Li (xhli@capitalbio.com)
Xixiong Kang (kangxx@VIP.sina.com)
Yongyong Shi (shiyongyong@gmail.com)
Yimin Sun (ymsun@capitalbio.com)

Version: 2 Date: 14 April 2012

Author's response to reviews:

Dear Editors:

We would like to resubmit the enclosed manuscript entitled “Association between Paraoxonase gene (PON) and stroke in the Han Chinese Population” after revisions, which we wish to be further considered for publication in “BMC Medical Genetics”. We quite appreciate your favorite consideration and insightful comments. Now we have revised the “MS: 2057366847070665” according to your comments, and found these comments are very helpful. We hope this revision can make the paper more acceptable. The revisions were addressed point by point below.

[1] We have tried our best to improve the quality of written English. However, if you find this revised version is still not satisfactory, we will ask a native speaker to correct the language again.

[2] All the participants gave their written informed consent. And we have added this description to the method part in the present version.

We deeply appreciate your helpful suggestions, and we look forward to learn more from you. If you have any queries, please don’t hesitate to contact me at the address below.

Thank you and best regards.

Yours sincerely,

Wenjin Li, wenjinliwl@gmail.com
Corresponding author:
Name: Yimin Sun, E-mail: ymsun@capitalbio.com;
Name: Yongyong Shi, E-mail: shiyongyong@gmail.com.